**Author’s response to reviews**

**Title:** Efficacy and safety of Guhong injection for treating the Coronary microvascular disease: study protocol for a randomized controlled trial

**Authors:**

Jieqin Zhuang (1004548400@qq.com)
Shuling Liu (1843397193@qq.com)
Hairong Cai (2587921581@qq.com)
Xingzhen Dai (451519983@qq.com)
Yanhong Chen (45151998@qq.com)
Zilin Jin (258792158@qq.com)
Bojun Chen (1198039070@qq.com)

**Version:** 1  **Date:** 05 Dec 2019

**Author’s response to reviews:**

Response to Reviewers:

Thanks to the reviewers for their detailed comments. I have highlighted all changes with red color. The details of our amendments one by one are listed as follows:

1. We have changed the description according to 1-6, 8, 9, 12, 13 points.
2. We have further elaborated on the blind method in detail according to 7, 8, 10, 11 points.
3. As for 9 point, the symptom in the original text mainly refer to angina pectoris, while TCM syndromes refer to a series of TCM symptoms and signs on coronary heart disease. We have made more accurate changes to the description.
4. As for 15, 16 point, we have added a detailed description about the part of interim analyses and sample size calculations in the article.
5. We have revised the for problematic references again.

Thank you very much for your careful review.